Author's response to reviews

Title: Prospective cohort study of the relationship between neuro-cognition, social cognition and violence in forensic patients with schizophrenia and schizoaffective disorder

Authors:

Ken O'Reilly (oreillk5@tcd.ie)
Gary Donohoe (DONOHUG@imx1.tcd.ie)
Ciaran Coyle (ciaran.coyle@hse.ie)
Danny O'Sullivan (danny.osullivan2@hse.ie)
Arann Rowe (arann.rowe@kcl.ac.uk)
Mairead Losty (mlosty@tcd.ie)
Tracey McDonagh (mcdonagt@tcd.ie)
Lasairiona McGuinness (lasairiona_mcguinness@hotmail.com)
Yvette Ennis (yvettetennis@eircom.net)
Elizabeth Watts (elizabeth.watts2@hse.ie)
Louise Brennan (louisepbrennan@eircom.net)
Elizabeth Owens (owensliz@hotmail.com)
Mary Davoren (davorem@tcd.ie)
Ronan Mullaney (rmullan@tcd.ie)
Mary Davoren (davorem@tcd.ie)
Zareena Abidin (zareenaabidin@yahoo.ie)
Harry G Kennedy (kennedh@tcd.ie)

Version: 6
Date: 24 June 2015

Author's response to reviews:

Dear Editors

while waiting for your response to our revised manuscript and responses to the referees, we have made some minor corrections to the text and added some references to the recently published work of Coid's group on the use of mediation analysis to describe the evidence for a causal link between delusions and violence that is mediated by anger. We have also corrected some minor graphical points in the additional material slides.

My colleagues and I hope to hear from you soon

With best regards

Harry Kennedy
Professor Harry Kennedy
Executive Clinical Director
National Forensic Mental Health Service
Central Mental Hospital, Dundrum, Dublin 14, Ireland
And
Clinical Professor of Forensic Psychiatry
Department of Psychiatry
Trinity College Dublin
Phone +35312989266
Caroline Higgins, PA +35312157458
Fax +35312157570
Please copy e-mails to Caroline.Higgins2@hse.ie